

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Cuna, Luigi

Working Paper Assessing Household Vulnerability to Employment Shocks: A Simulation Methodology Applied to Bosnia and Herzegovina

Quaderni - Working Paper DSE, No. 528

Provided in Cooperation with: University of Bologna, Department of Economics

Suggested Citation: Cuna, Luigi (2004) : Assessing Household Vulnerability to Employment Shocks: A Simulation Methodology Applied to Bosnia and Herzegovina, Quaderni - Working Paper DSE, No. 528, Alma Mater Studiorum - Università di Bologna, Dipartimento di Scienze Economiche (DSE), Bologna,

https://doi.org/10.6092/unibo/amsacta/4762

This Version is available at: https://hdl.handle.net/10419/159369

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by-nc/3.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Assessing Household Vulnerability to Employment Shocks: A Simulation Methodology Applied to Bosnia and Herzegovina

Luigi Cuna Post Doctorate Fellow Dipartimento di Scienze Economiche Università degli Studi di Bologna <u>cuna@spbo.unibo.it</u>

Draft for Comments

October 2004

Abstract

Household vulnerability relates to the incapacity of a household to preserve its welfare because of negative events. In Bosnia and Herzegovina, vulnerability is a central issue in the policy debate: many households are at risk of poverty due to fragile livelihood systems or high exposure to shocks. Since no panel data are available in Bosnia and Herzegovina, a micro-simulation methodology is adopted on the basis of a household consumption model based on quintile regression. Shocks are simulated in the labour market and the methodology consents the identification of the typologies of households which suffer from severe welfare losses or that are plunged into poverty after the shocks. The results show that the vulnerable households identified change when different shocks or a diverse definition of vulnerability are taken into consideration. At the same time, the characteristics of households vulnerable to poverty are shown to be different with respect to those of the poor.

JEL Classification: C15; C31; I32; D31; J60

Key-words: Bosnia and Herzegovina, Micro-simulation; Vulnerability; Welfare Dynamics; Labour Market Shocks, Living Standards Measurement Survey.

1. Introduction

In general, the concept of vulnerability relates to the occurrence of events which negatively impact on something, such as individuals, households, communities, countries, enterprises. The analysis presented in this paper focuses on households and defines vulnerability as the inability of a household to secure its living standards in the face of a certain negative event. Vulnerability is therefore the combination of two facts: the exposure to a negative event and the capacity of the households to cope with it (Chambers, 1989). Following on this definition, an assessment of vulnerability should examine the nature of the shock (Sinha and Lipton, 1999), how this is transmitted to the household as well as the coping mechanisms available at the household or community level (Holzmann et al., 2001; Moser, 1996, 1998; Shaffer, 2001; Dercon, 2001). At the same time, a methodological criterion is needed for identifying the households unable to secure their living standards in the face of a negative event¹. The literature proposes two main approaches to household vulnerability.

In the first approach, vulnerability is defined as a downturn in consumption which can be attributed to an "uninsured exposure to risks" (Hoddinott et al. 2003) or, more generally, to the lack of effective coping mechanisms. Glewwe and Hall (1998) provide one of the earliest applied analyses of household vulnerability defined as variability in household consumption. They identify the socio-economic groups experiencing positive or negative changes in their consumption value between 1985 and 1990². The literature has also analysed vulnerability by looking at the covariance between income changes and consumption changes (Amin et al. 2001). Vulnerability is estimated by looking at the response of household consumption to household-specific (*idiosyncratic*) income shocks. Higher covariance means higher vulnerability of consumption to income risk. The application of this methodology to the participants of micro-credit programmes in Bangladesh shows that households below the poverty line have significantly higher average vulnerability than those above the poverty line.

This approach provides evidence of the incapacity of existing risk coping mechanisms to protect household consumption from income fluctuations (Hoddinott and Quisumbing, 2003). However, this literature does not provide information on *what* creates vulnerability. This is the focus of Dercon and Krishnan (2000): consumption fluctuations are linked to shocks experienced by rural households. The authors analyse movements of household consumption along an 18 month dataset in Ethiopia. The authors find that shocks have a negative effect on household welfare: harvest failure is the most frequently quoted cause of hardship; other problems are related to public policy, labour and livestock³.

¹ It is important to point out that this paper analyses household living standards by exclusively looking at the value of goods and commodities consumed by the household. This approach can be considered too narrow with respect to the broad and multidimensional nature of household well-being but it is in accordance with most of the literature on poverty and vulnerability (Chaudhuri, 2001; Amin et al., 1999; Prichett et al. 1998).

 $^{^{2}}$ Morduch and Gamanou (2002) argue that the change in consumption over two observations cannot constitute a reliable measure for vulnerability and few data sets have a long enough time dimension to yield a reliable measure of variability for each household over time.

³ The authors also found that seasonal effects are highly significant in determining fluctuations in household consumption. In particular, households boost consumption in peak periods and reduce in response to seasonal price movements. This finding is in contrast with the hypothesis of consumption smoothing.

The second approach to vulnerability looks at the decline in living standards below a certain threshold, such as the poverty line⁴. In both approaches, vulnerability manifests itself in terms of downward fluctuation in living standards. However, the second approach exclusively looks at the households whose decline is such that the poverty line is crossed. This definition of vulnerability excludes those households among the non-poor who are well- enough off so that when they experience the decline in welfare provoked by the shock they remain non-poor.

Within this framework, various methodologies have been proposed in literature. Chaudhuri et al. (2001) define vulnerability as the ex-ante risk of facing poverty in the future, conditional on the current observable household characteristics. They estimate the probability that a currently non-poor household will fall below the poverty line and the probability that a currently poor household will remain in poverty even if it does not experience any large adverse welfare shock⁵. Similarly, Christiansen et al. (2000; 2001) define vulnerability as the probability statement regarding failure to attain a certain threshold of well-being in the future. The methodology elaborated in Christiansen et al. (2000; 2001) requires the definition of the time horizon (set by the authors at one period ahead) and of the indicator of well-being (consumption). The household's ex-ante distribution of future consumption is obtained from a flexible heteroskedastic regression specification, which allows the prediction of the ex-ante mean and variance of future consumption for each household based on its current characteristics and assets. The methodology also requires the definition of the poverty line and of a probability threshold such that a person or household is considered vulnerable if its probability of shortfall exceeds the given threshold. Prichett et al. (1998) instead expand the analysis on a multiple time horizon and they define vulnerability as the probability of observing at least one episode of poverty in the future.

However defined, the identification of vulnerable household is a very delicate task. Because of the intrinsic uncertainty that features human life, it is impossible to draw definitive conclusions on the households which are at risk of falling below the poverty line or at risk of experiencing welfare downturns in the future. Nobody can predict what it is going to happen in the future: the typology, the size, the persistency and the severity of the negative event are unknown. Similarly, the household reaction and the capacity to cope with the shock are unpredictable. Most of the methodologies (Chistiaensen et al, 2000; Prichett et al, 1998;

⁴ The attention towards households experiencing decline into poverty is not new in economic and social history. Looking back at the mid-Thirteenth century, the almsgiving confraternities and the assistance institutions included among their concerns the shamefaced poor, at that time called the *vergognosi*, the Italian word for shamefaced (Pullan, 1999). This term encompassed almost all kinds of people in distress (such as decayed gentlefolk, craftsmen, shopkeeper, failed merchants, widows of wealthy husbands, journeymen, etc.) who had scruples about begging, hence defined as *vergognosi*. A similar concern can be found in France during the Second Empire, when the term *déclassé* was used to indicate those previously wealthy individuals who fell into a lower social class (Ricci, 1999). In contrast with the *vergognosi*, who suffered from an individual or household based type of negative happening, the emergence of the *déclassé* was mainly provoked by the rapid revolutionary changes experienced in French society. Both the *vergognosi* and the *déclassé* were incapable of preserving their earning capacity and economic status in front of the changes occurring in their lives or in their society.

⁵ It must be specified that Chaudhuri et al (2001) utilise a cross-section survey. The utilisation of a cross-section survey forces the establishment of very stringent hypothesis regarding the process generating household consumption. The authors assume the consumption generation process in which "a mean-zero disturbance term (...) captures idiosyncratic factors (shocks) that contribute to different per capita consumption levels for households that are otherwise observationally equivalent". "The idiosyncratic factors that contribute to different per capita consumption levels are identical and independently distributed over time for each household". And "the uncertainty over future consumption "stems solely from the uncertainty about the idiosyncratic shock, e_{h} , that a household will experience in the future".

Glewwe et Hall, 1998) proposed for measuring household vulnerability identify vulnerable households when they have already fallen into poverty or when they have already experienced welfare downturns. By doing so, assessing vulnerability does not lead to the identification of today's households at risk of experiencing poverty, but of those who *were* at risk of poverty who *are* currently in a poverty status.

This paper attempts to address these shortcomings by adopting a simulation methodology. In this way the occurrence of shocks can be modelled and their impact on household living standards explored. Micro-simulations are often utilised for evaluating the impact of development policies on poverty (Datt et al., 2000; Mukherije and Benson, 2003; Wodon, 2003) but their application to vulnerability assessment is quite unexplored.

The paper proposes a quintile regression for estimating a household consumption model (Koenker and Bassett, 1978; Buchinsky, 1998). A quintile regression allows hypothesising that the shock might have a diverse impact on household consumption depending on the relative position of the household in the welfare distribution. Two types of shock are simulated: the first concerns employment loss in the formal public sector; the second in the informal dependent one. The outcomes of the shocks are analysed in terms of both downturn in household welfare and poverty increase. Two different facets of vulnerability are examined: "vulnerability to welfare loss" that defines the households experiencing a downward movement by a minimum amount of two deciles and "vulnerability to poverty" which looks at the households which fall into poverty after the shock. This methodology produces relevant policy implications since it allows identifying which households categories should be regarded as in need of targeted social support. At the same time, the first three FGT measures of poverty after the occurrence of the shock⁶.

The paper is organised as follows: Section 2 presents the general poverty and vulnerability situation in Bosnia and Herzegovina; Section 3 describes the dataset utilised and the methodology adopted for modelling household consumption and estimating vulnerability; Section 4 introduces the variables utilised in the regression model; Section 5 reviews the results of the regression; Section 6 analyses the results of the vulnerability simulations based on a shock which occurred in the formal public sector; Section 7 analyses the impact on vulnerability of a shock occurring in the informal sector; finally, Section 8 concludes.

2. Poverty and Vulnerability in Bosnia and Herzegovina

Bosnia and Herzegovina is one of the poorest countries in Eastern Europe. The country Human Development Index (HDI) is 0.718, 27% lower than the EU average (0.914). In comparison with other Eastern European countries, the Bosnia HDI outperforms only

$$P_{\alpha} = \frac{1}{N} \sum_{i=1}^{P} \left(\frac{z - c_i}{z} \right)^{\alpha}$$

⁶ The headcount, the poverty gap, and the squared poverty gap are the first three measures of the FGT class of poverty measures. The general formula for this class of poverty measures depends on a parameter α , which takes a value of zero for the headcount, one for the poverty gap, and two for the squared poverty gap in the following expression (Foster, Greer, and Thorbecke 1984):

Moldova (N°105, 0.699). Until recently, no data were available for poverty analysis. The first quantitative analysis of household poverty in Bosnia and Herzegovina is proposed by Bisogno and Chong (2002). The authors identified a relative poverty line using per capita income as well being indicator⁷. According to this study, in Bosnia and Herzegovina 27.3% of the population live below the general poverty line (set at 60% of the median per capita income) and 11.5% under the extreme poverty line (set at 30%).

The Living Standard Measurement Survey (LSMS) established for the first time a "properly researched and calculated absolute poverty line"(UNDP, 2002, p.49) which amounts to KM (Konvertible Marka) 2,198 per person annually; on the other hand the extreme poverty line in this country amounts to KM 760 per person yearly. Household consumption was chosen as indicator of the household well-being. According to the estimates, no-one is living in extreme poverty line (LSMS, 2001; PRSP, 2002)⁸. At the same time, significant territorial differences in poverty incidence emerge: the Federation of Bosnia and Herzegovina (FBiH) registers a poverty headcount of 15% compared to 25% of the Republic Srpska (RS)⁹.

In Bosnia and Herzegovina, together with the poverty figures, terms like "vulnerability" and "risk of poverty" often recur in the social policy debate. The Poverty Assessment believes that "a sizeable proportion of the population faces the risk of falling into poverty". Therefore "the percentage of population classified as poor on the basis of the current consumption levels is only a fraction of those in Bosnia and Herzegovina who must worry about, and struggle to avoid becoming poor at some point in the relatively near future" (World Bank, 2003, p. 2). There is something more than the official poverty figures to be said on the household exposure to shocks, their weak endowment of productive assets, the uncertainty of income, etc¹⁰.

In the country PRSP, households are defined vulnerable whose per capita consumption is "slightly above the poverty line". A vulnerability line, set at 50% above the general poverty line, is drawn and a household is defined vulnerable if its per capita consumption falls in the interval determined by the two lines. This approach is highly

⁷ This decision poses severe methodological problems since income is not the best indicator for measuring living standard (Deaton and Muellbauer, 1980), especially in Easter Europe, where income volatility and unreliability of income data tend to be very high (Grootaert and Braithwaite, 1998). Moreover, the dataset utilised by Bisogno and Chong scarcely involved remote communities because of transportation problems and security reasons. The exclusion of these areas clearly affects the final results especially in terms of underestimation of poverty levels.

⁸ It must be noticed that in the initial LSMS publications (PRSP, 2002 cited also in UNDP, 2002; in Ardeni and Andracchio, 2003 and Cuna and Alasia, 2003) the KM 1843 general poverty line and KM 747 absolute poverty line were utilised. On the basis of these values, 19% of the population in Bosnia and Herzegovina was found below the general poverty line. The successive publications based on the LSMS data (BHAS, 2002; Poverty Assessment, 2003) develop a higher poverty line (set at KM 2189) associated with the same poverty headcount (19%). In order to be consistent with the poverty estimates proposed by international and national authorities, the elaborations conducted in this research are based on the KM 1843 poverty line.

⁹ The Federation of Bosnia and Herzegovina (FBiH) and the Republic Srpska (RS) have been created in 1995 after the Dayton Peace Agreement. The RS comprises 49% of the territory of Bosnia and Herzegovina and it reflects the area controlled by Serb during the war; the FBiH constitutes 51% of the overall territory and corresponds to the area controlled by the Muslim-Croat alliance.

¹⁰ A research contribution related to household assets endowment is proposed by Ardeni and Andracchio (2003). Following on the Amartya Sen's capability approach (Sen, 1993), the authors found that the deprivation in Bosnia Herzegovina is higher than consumption poverty.

intuitive: the identification of the vulnerable households is simply made by counting the number of households falling in the pre-determined interval. However, the counting of the households whose living standard is slightly above the poverty line is a measure of the welfare distribution rather than household vulnerability¹¹. This way of looking at vulnerability identifies the vulnerable without explaining *why* they are vulnerable. In other words, why should policy-makers look at those households who are slightly above the poverty line? Why does being slightly above the poverty line matter? These questions are of central importance because households defined as vulnerable according to the PRSP approach, might count on a quite secure livelihood system although proximal to the poverty line.

World Bank (2003) integrates the approach described above with the analysis of employment risk, which is considered as "the leading and most significant factor affecting household well being" (World Bank, 2003)¹². For determining employment risk, unemployment rate is considered the "most direct measure of the risk of employment loss: for someone who is employed now, a high unemployment rate today implies that it would be difficult to find a new employment in case the current one terminates" (World Bank, 2003, p. 21). According to the PRSP figures, the unemployment rate in Bosnia and Herzegovina is 16.4% (16.9% in the FBiH and 15.8% in the RS)¹³; 14% of the households have "at least one unemployed member, suggesting that the risk of unemployment is evenly spread in the society". In particular, "poor households seem to have higher incidence of unemployment, but even non-poor households look susceptible to this threat" (World Bank, 2003, p. 21). However, "contrary to the widespread belief, unemployment risk does not affect the majority of the population" (ibid., p.22). Hence, it cannot be concluded that the unemployment figures raises special vulnerability concerns.

By limiting the analysis of household exposure to shock to employment issues, unemployment risk can reasonably be considered a more complicated issue with respect to the analysis proposed by the Poverty Assessment. Workers from diverse sectors or endowed with diverse skills experience risk of dismissal for various reasons (privatisation, macroeconomic downturns, etc) and at various degrees. Also, in order to assess household vulnerability, the analysis of unemployment risk must be combined with the analysis of the household livelihood sources: the dismissal of an income-earner may severely impact on households made-up of a single earner, whereas it may have a lower impact on multiple income source households.

¹¹ For example, one may be interested in understanding the shares of population living within various welfare intervals. Interesting results may also emerge if the proportion of households living in the two intervals (one between zero and the poverty line, the other between the poverty and the vulnerability line) is monitored across various periods.

poverty line, the other between the poverty and the vulnerability line) is monitored across various periods. ¹² The survey of risk factors provided by the Poverty Assessment also regards health risk, corruption and political governance risk.

¹³ These values are calculated by taking into consideration the individuals without work, willing to work and actively looking for a job and ready to take one. Following on alternative definition, such as taking those registered as unemployed, the unemployment rate rises to 40%. These estimates contrast with the calculations made by UNDP (2000) that put the unemployment rate at 43% in the FBiH and 40% in the RS. Other calculations made by taking into consideration the presence of employed in the informal sector decreases the estimate at 22.2% for Bosnia and Herzegovina, 23.3% in the RS and 21.6% in the FBiH (UNDP, 2002). These values are "broadly in line with those of other Balkan countries such as Bulgaria, Croatia and Yugoslavia" (Poverty Assessment, p. 21); in comparison with OECD standards "these rates are very high, but they are lower than the turbulent history of the country and popular belief suggest" (UNDP, 2003).

3. Data and Methodology

The data utilised in this paper are drawn from the Living Standard Measurement Survey (LSMS). This survey was carried out in Bosnia in the autumn of 2001 and covers a sample of 5,400 households¹⁴. The sample is constructed following a probabilistic methodology that implies the possibility to attribute a confidence interval to each estimate made on the basis of the sample observations. In general, the sample design for any survey requires information on the universe of the households and the individuals in the country (usually coming from a census or administrative records). In Bosnia and Herzegovina, the most recent census dates back to 1991: the data contained in this census are obsolete due to the simple passage of time but, more importantly, also due to the massive population displacements which occurred during the war. Therefore, at the initial stages of the LSMS, a master sample was constructed. This was based on a selection of municipalities and a full enumeration of the selected municipalities. The 146 municipalities of the country were grouped into three strata, Urban, Rural and Mixed, within each of the two Entities, on the basis of the information contained in the 1991 census¹⁵. Municipalities were selected with probability proportional to estimated population size within each stratum. Later, households were selected from the municipalities with the same criteria.

The methodology adopted in this paper refers to micro-simulations. This choice is due to the fact that panel data for households are currently not available in Bosnia and Herzegovina. On the other hand, a cross-section dataset does not permit the monitoring of the proportion of households experiencing welfare downturns due to shocks. Micro-simulations are based on multivariate regression techniques which permit the estimation of the importance of various factors to a certain outcome and, consequently, predict the effects of change in selected variables. In other words, "having estimated a consumption model, (...) it is possible to generate simulations to predict the reduction or increases in general poverty levels that results from unit change in selected household or community characteristics" (Mukherjee and Benson, p. 351, 2003). Midhet et al. (1998) and Christiaensen and Alderman (2001) applied micro simulation techniques to the phenomenon of maternal mortality in Pakistan and child malnutrition in Ethiopia. Wodon (2002) recently developed a software based application that examines poverty reduction targets based on micro-level analysis. Micro simulations are utilised for measuring the effects of alternative policies on poverty indicators (Datt et al. 1998; Datt et al., 2000; Mukherjee and Benson, 2003) or to construct spatially-disaggregated poverty maps (Lanjow et al. 2002).

¹⁴ The information of the dataset is based on LSMS, (2001). "Basic Information Document - Bosnia and Herzegovina.

¹⁵ A municipality was defined as urban when 65 percent or more of the households considered themselves as urban, whereas rural municipalities are those where the proportion of urban households is below 35 percent. The remaining municipalities were classified as "Mixed" Municipalities. Brcko was excluded from the sampling frame.

In formal terms, this methodology is based on a household welfare model that allows us to estimate how a particular characteristic impacts on household welfare conditional on other variables. The model is specified as

$$\ln c_i = \alpha + \beta x_i + \varepsilon_i$$

where j=1,2,...n is the unit of analysis, namely the household, x_j is a vector of exogenous determinants of household welfare, α is the intercept, β is the set of coefficients to be estimated by the regression and ε_j is the random error term.

Ordinary least squares regression is the econometric model generally adopted for computing the regression coefficients (Glewwe, 1991). One of the limitations of this model is that the marginal effect of each variable is constant, equal to the estimated coefficient, along all the household welfare distribution. Hence, the model might be considered inadequate to capture the possibility for a factor to verify diverse values among different groups of population. It might be the case that the elasticity of household welfare to a certain variable differs as one moves along the welfare distribution and that some characteristics can be significant only for some groups of households.

By taking into consideration this shortcoming, the paper adopts a quintile regression (Koenker and Bassett, 1978) that consents to determine "whether a household's position in the expenditure distribution affects differentially the relationship between expenditure and household characteristics" (Anderson and Pomfret, 2000, p. 509). In other words, quantile regression relaxes the constraint imposed by ordinary least square estimation and it allows "the effect of the explanatory variable to vary at different points in the expenditure distribution, permitting the determinants of per capita expenditure to differ between rich and poor households" (Anderson and Pomfret, 2000, p. 510).

In quantile regression, the linear estimation does not describe $E(y_j|X_j)$, as in ordinary least square, but $f(y_j|X_j)$, for a quintile q of the distribution where 0 < q < 1. To obtain an estimate for quantile q, the values of $y_j - X_j\beta$ at the estimated value of β are weighted; if a residual is negative, it is weighted -(1 - q), and if residual is positive, it is weighted q. As commented by Anderson and Pomfret, (2000), "whereas an OLS regression provides the best linear estimator of the mean, quantile regression offers the best linear approximation of the effect of explanatory variables at various quantile of the dependent variable" (Anderson and Pomfret, 2000, p. 510).

This paper present three models: they refer to the 25th, 50th and 75th quantile. In each regression, a vector of parameters $\hat{\beta}$ is estimated for the set of explanatory variables x_1 , x_2, \ldots, x_n . In order to take into account the difference in poverty and determinants of living standards across the Bosnia and Herzegovina territory, the analysis separately refers to the RS and the FBiH.

The predicted per capita consumption level \hat{c}_{j} for any household j can be expressed as:

$$\hat{c}_{j} = e^{(\hat{\beta}' x_{j})}$$

Since three regressions are developed, three estimates of consumption can be calculated for each household. A criterion is therefore needed for developing a unique vector of estimated household consumption level. The analysis takes into consideration the distribution of natural logarithm of household consumption at Entity level. Two cut-off points are identified for dividing the population in three deciles. The 25th quantile regression estimates are utilised for the population falling below the first cut-off point, the 50th quantile regression for the population falling between the first and the second cut-off point, the 75th quantile regression for the population having household consumption above the second cut-off point¹⁶. The predicted consumption level of household *j* is used as benchmark value for attributing the "before the shock" poverty status¹⁷. A household is therefore classified as poor if $\hat{c}_i < z$ (where z is the poverty line).

As commented above, micro-simulations are based on the idea that changes in regression independent variables can be interpreted as outcome of development policies (Datt et al., 1998; Datt et al. 2000) or of negative events. In this paper, the micro-simulations are based on shocks occurring in the household employment characteristics. The effect of the shock on household consumption (predicted) after changes occurred in the x_i variable depends on the estimated value of $\hat{\beta}_i$. For households below the first cut-off point, the elasticity of \hat{c}_j to x_i is calculated on the basis of the 25th percentile model; for households between the first and the second cut-off point, the 50th percentile regression model is used; for households above the second cut-off point, the 75th regression estimates are utilised¹⁸.

The after-shock predicted consumption \tilde{c} is used for attributing the welfare status of household *j* after the occurrence of the shock; if $\tilde{c}_j < z$, the household is defined as poor. Confronting the poverty status before and after the simulation, three typologies of households can be distinguished: the "always poor" households that record $\hat{c} < z$ and $\tilde{c} < z$, the "vulnerable to poverty" households having $\hat{c} > z$ and $\tilde{c}_j < z$ (hence, they become poor after the occurrence of the shock) and the never-poor households ($\hat{c} > z$ and $\tilde{c}_j > z$).

In addition, the "before the shock" and "after the shock" welfare distribution can be compared. This is made by employing transition matrices. Transition matrices show how households living in a certain population decile in a given period move to other deciles in the subsequent period (Baulch and McCulloch, 1998). In this paper the before-the-shock welfare distribution (based on predicted value of household consumption) and the simulated after-the-shock one are compared. On the basis of the transition matrix, "vulnerable to welfare losses" are defined those households which lose more than two deciles in the welfare ranking after the shock (Haughton et al.; 2001).

Before presenting the results of the paper, two important caveats must be pointed out. As already mentioned, the shock model hypothesises employment losses among Bosnia and Herzegovina household. It should be considered that when an individual loses his job, the household will undertake actions that are not modelled in the regression. This is the case for example of other household members that may decide to work, or to work additional hours;

¹⁶ The values of the cut-off point correspond to 7.62 and 8.03 for the RS and 7.78 and 8.14 in the FBiH.

¹⁷ Needless to say, *predicted* household consumption may differ from the *real* household consumption. Using predicted consumption for assessing household poverty may therefore lead to error of inclusion (households who are non-poor are counted as poor) and exclusion (household who are poor are counted as non-poor). On this issue, the literature considers "incorrect (...) to compare the actual consumption and poverty levels derived through the poverty analysis of the survey with the predicted levels derived using the determinants of poverty model" (Mukherjee and Benson, p. 351, 2003).

¹⁸ When an OLS regression is used, the impact of the shock on household consumption is equal for all the households in the welfare distribution. This implies that loss of employment has the same impact (measured in terms of loss in welfare) for both poor and rich households. The adoption of a quintile regression consents to overcome this shortcoming.

similarly, the households may have access to informal support from neighbours or families living abroad, etc. In the model, no coping mechanisms are hypothesised to occur. Also, "(some of the) unobservable or unmeasured variables may also be key driving factors (of the modelled variable); (therefore) their omission may result in a bias of the estimated coefficients and the related policy simulations" (Christiaensen et al. 2002, p.145)¹⁹. A related shortcoming concerns the fact that the simulations assume that changes in the determinant variable do not affect model parameters and that other variables in the model are not affected although feedback mechanisms may operate in the reality.

The second caveat relates to the attempt to measure a dynamic phenomenon, such as welfare downturns, from a "static" regression. It is common opinion among economists that elasticity computed from cross-section data is different from those of time-series. Alasia and Soregaroli (2002) review the econometric debate around this shortcoming in their work on determinants of food demand in Mozambique. Based on Khu (1959) and Houthakker (1965), the authors remind us that cross section estimates could not be a good proxy of the dynamic relation between the independent and the dependent variables. By looking at the response of consumption to income changes, Gardes et al. (1996) conclude that "elasticity computed using cross-section data cannot be used to predict change in consumption over time".

A final issue regards the presence of incomplete observations. Datt et al. (2000) and also Mukherjee and Benson (2003) include the observations with missing data by developing a set of dummy variables that records the value of one if the household is missing data for a certain variable. According to the authors, this method consents not to exclude the households recording missing values for some variables but non-missing responses for other. However, for the parts of the LSMS utilised in the regression, missing values are not numerous. It is therefore decided to adopt a very simple treatment in which households verifying a missing value in any of the variables are excluded from the regression. The exclusion of households with missing values produces an aggregate loss of less than 2% of total observations.

4. The Variables²⁰

<u>Household Consumption</u>. The value of per capita household consumption is the dependent variable of the regression model²¹. Per capita household consumption includes food and non-food items, as well as own production, durable goods and housing expenditures. The adoption of a per-capita basis is motivated by willingness to keep the research coherent

¹⁹ In the Bosnia and Herzegovina LSMS, several potential determinants of household welfare are not included or poorly explored. For example, the remittances received from families living abroad is not examined; also, in case of agricultural households, the Bosnia LSMS is inadequate to capture issues related to land productivity and harvest losses.

²⁰ The descriptive statistics are reported in appendix.

²¹ Datt (2000) identifies four considerations in favour of using consumption instead of income as indicator of household welfare. First, income can be interpreted as a measure of welfare *opportunity* while consumption is interpretable as a measure of welfare *achievement* (Atkinson, 1989). The second relates to the permanent income hypothesis which suggests that consumption typically fluctuates less than income. The third consideration concerns the difficulties in measuring income (Deaton, 1997). The final consideration is based on the idea that income is not an insightful measure of welfare for households whose income comes from self-employment activities (Deaton, 1997; Datt et al., 2000). These concerns are very relevant in Bosnia and Herzegovina, where many people are paid very irregularly, with several months of wage arrears being common.

with the practices adopted both in the Bosnia and Herzegovina PRSP and in econometric poverty analysis, where per capita consumption is preferred to adult-equivalent measures (Mukherjee and Benson, 2003; Datt et al., 2000; Datt & Joliffe, 1999). Choosing per capita consumption as welfare indicator instead of adjusting per adult equivalent implies that "(i) everyone in the household receives an equal allocation of items consumed irrespective or age and gender; (ii) everyone in the household has the same needs irrespective of age and gender; (iii) the cost of two or three or more people living together is the same as if they lived separately" (Skoufias et al., 1999, p. 77).

For the independent variables, special care was given to select variables which are not affected by the current level of household welfare. In other words, the model includes variable "whose values were determined outside the current economic system of the household, but which determine the current level of welfare" (Judge et al., 1985).

<u>Demographic characteristics</u>. This set of variables includes the characteristics of the household in terms of size, composition and displacement status. Three categories are identified according to the age of the household members: below 8 years old, 8-14 years old, 15-59 years old. For individuals aged between 15 and 59, a gender distinction is applied. The household size is introduced in a log form. The regression also explores the connection between household welfare and the gender of household head. A dummy variable is introduced to indicate whether the household head is a temporary resident (e.g. a displaced person)²².

Education. The regression includes eight variables indicating different education attainments of household members older than 15 years old (called adults for simplicity). This is because the investment in education made by household adults cannot be influenced by the current household living standards because it dates back to years ago. The first education variable in the regression considers the number of adult members over total household size who never attended school and unable "to read and write with understanding, short, and simple sentence during everyday life" (LSMS, 2001). The second variable identifies the adult members who never attended school or never attained qualifications but that are capable of reading and writing. The investment in human capital and education is analysed by looking at the number of adults with primary, secondary and higher level of education²³. In line with the most recent econometric analysis of household welfare (such as Datt et al., 2000), the education variables are differentiated by gender.

<u>Employment</u>. This set of variables refers to household employment and livelihood characteristics. The presence of unemployed is represented by the number of unemployed over total adults in the household. The case of individuals occupied in informal labour is represented by two variables. The first includes members who are small scale farmers on own farm, unpaid individuals who support other members in their work and those who perform other activities such as sale of agricultural goods and other unspecified services. These are defined as generic informal occupation. Following on Vodopivec (2002), an additional category of informal labourers is included. These are defined "informal dependent

²² The displaced persons are "considerably more susceptible to poverty than the population that was not forced to move" (PRSP, 2002, p.10): they face in fact "greater difficulties in finding a job and often forced to accept jobs that other groups were not interested in" (ibid., p.11). The choice to exclusively look at the displacement of household head is due to the high number of missing values in the LSMS module for migration.

²³ The latter includes high school, university, master or doctorate.

employees" and are constituted by persons working for private enterprises but who are not fully registered, namely do not receive pension contributions. This categorisation is in accordance with Bernabè (2002) which proposes a distinction between unregulated and unregistered activities whose main purpose is to meet basic needs and activities which are concealed to avoid taxes and regulations. These two aspects are very relevant in a country such as Bosnia and Herzegovina, where enterprises attempt to escape registration and legislation procedures while households and individuals turn to small scale income activities to generate livelihoods in absence of formal employment opportunities (World Bank, 2003). Formal labour is described through three variables: the first identifies household members working in public enterprise, institutions and organisations. The second identifies those working for employers in the private sector but who regularly receive pension contributions. The last variable represents the owners or co-owners of enterprises or small-businesses. A variable is derived to represent the number of pensioners over total household size. Finally, a variable represents the size of land cultivated by the household, regardless of the property status. This variable captures both the fact that the household is engaged in agricultural activity and the scale of this activity.

Location. In the literature, the location dimension is usually captured by introducing a distinction between rural and urban areas. Following on Cuna and Alasia (2003), in Bosnia and Herzegovina, the use of territorial dummies based on the classification of municipalities by regional types (rural/mixed/urban) as proposed by the LSMS does not adequately capture the territorial variation in poverty across the country. The set of location variables is therefore represented by 25 dummies, corresponding to each of the municipalities covered in the survey²⁴. In the regression, the excluded variables refer to Banja Luka in the RS, and to Centar municipality in the FBiH²⁵.

5. Regression Results

Since the dependent variable is in natural logarithm form, the estimated regression coefficients measure the percentage change in per capita consumption within the household resulting from a unit change in the independent variable. The values of the "pseudo" R-square index are encouraging: for the RS, the 25th quantile regression has 0.26 pseudo R-Square; 0.27 for the 50th quantile and 0.28 for the 70th quantile. For the FBiH, the pseudo R-Square are 0.30, 0.32 and 0.34 respectively. All the sets of variables are jointly significant at the 1% confidence level in the two equations in both Entities. The regression is developed through a simultaneous quantile regression model. Standard errors are obtained using bootstrapping resampling (Efron and Tibshirani, 1993; Wu, 1986), setting the random seed number at 100.

<u>Demographic Characteristics.</u> The F-test demonstrates that the demographic variables are jointly significant at 1% confidence level in both Entities. The regression evidences the strong and negative relationship existing between household size and value of per capita

²⁴ A similar approach was adopted by Ravaillon, M., and Wodon, Q., (1997). "Poor areas or Poor People", World Bank.

²⁵ Unfortunately, the Bosnia and Herzegovina LSMS does not include a community questionnaire. The availability of this dataset would allow inserting in the regression variables referring to infrastructure endowment and other relevant characteristics of the municipality in which the household lives.

consumption: the more numerous is the family, the lower its welfare. This finding confirms the PRSP concern for the high exposure to poverty of children living in large households.

Quantile Regression Results											
	R	epublic Srps	ska		Federation						
Variable Name	25 th	50 th	75 th	25 th	50 th	75 th					
	Quantile	Quantile	Quantile	Quantile	Quantile	Quantile					
Constant	8.079**	8.560**	9.062**	7.917**	8.415**	8.792**					
Persons 0- 7 years old	008	.003	.033	.029	0.004	001					
Persons 8 -14 years old	.028	.049	.066	012	008	009					
Females 15-59 years old	001	.025	.061**	.065**	.055**	.056*					
Males 15-59 years old	000	.026	.019	.043	.044**	.061**					
Log of household size	369**	470**	571**	581**	615**	671**					
Dummy, 1 if hh head is temp. resident	134**	116**	079**	.067	084**	099**					
Dummy, 1 if hh head is a women	.030	024	024	.040	.018	011					
Age of household head	003	011	018**	.012	.008	.006					
Square of age of nousenoid head	000	.000	.000*	000	000	000					
Adults unable to read/write on hh size	259*	278**	122	242**	221**	271**					
Ad. with no qualifications on hh size	.245	.212	.260	.051	.064	.028					
Male ad. with primary school on hh size	.331	.328	.423*	.147	.093	006					
Fem. ad. with primary school on hh size	.285	.247	.226	.050	.032	039					
Male ad. with secondary school. on hh size	.483	.470*	.577**	.357	.368*	.245					
Fem. ad. with secondary school on hh size	.600*	.470*	.471*	.316	.271	.146					
Male ad. with higher education on hh size	.789**	.768**	.854**	.708**	.671**	.627**					
Fem. ad. with higher education on hh size	.747*	.627**	.653*	.466*	.427**	.461*					
N° of pensioners on hh size	.085	.107*	.033	055	071	088**					
N° of unemployed on hh adults	125**	117	145*	218**	206**	128*					
N° of entrepreneurs on hh adults	.706**	.626**	.592**	.690**	.691**	.787**					
N° of inf. dep. employees on hh adults	.075	.145*	.107	.160	.090	.120*					
N° of inf. gen. employees on hh adults	.041	.092	082	049	002	010					
N° of formal public employees on hh adults	.227**	.205**	.119*	.311**	.286**	.297**					
N° of formal private employees on hh ad.	.296*	.379**	.119*	.283**	.318**	.434**					
Size of the land used	0.035	.034*	.067**	.000	000	001					
Dummy, 1 if living in Bania Luka	Omitted	Omitted	Omitted								
Dummy, 1 if living in Srpska Ilidza	152**	261**	258**								
Dummy, 1 if living in Cajnice	139	210**	185								
Dummy, 1 if living in Novi Grad (RS)	119*	119*	161								
Dummy, 1 if living in Prijedor	383**	348**	333**								
Dummy, 1 if living in Modrica	.054	016	038								
Dummy, 1 if living in Vicegrad	429	546**	520**								
Dummy, 1 if living in Knedzevo	.383**	246**	344**								
Dummy, 1 if living in Samac	001	009	022								
Dummy, 1 if living in Zvornik	.062	.010	025								
Dummy, 1 if living in Srbac	018	.065	.136**								
Dummy 1 if living in Centar				Omitted	Omitted	Omitted					
Dummy, 1 if living in Novigrad (Fed)				-151**	- 142**	- 178**					
Dummy, 1 if living in Novi Sarajevo				- 070	124**	- 182**					
Dummy 1 if living in Zenica				425**	- 487**	- 504**					
Dummy, 1 if living in Tuzla				240**	.266**	263**					
Dummy, 1 if living in Vogosca				051	.111**	153*					
Dummy, 1 if living in Travnik				078	108**	134*					
Dummy, 1 if living in Visoko				245**	304**	337**					
Dummy, 1 if living in Breza				258**	404**	400**					
Dummy, 1 if living in Zavidovici				119**	213**	246**					
Dummy, 1 if living in Gradacac				008	046	.017					
Dummy, 1 if living in Posusje				.162**	.119*	.042					
Dummy, 1 if living in Kakanj				194	243**	156*					
Dummy, 1 if living in Grude				.077	.205**	.089					

- Table 1 -

Dependent Variable: Natural Logarithm of Per Capita Household Consumption.* p<0.05; ** p<0.01.

In both Entities, the value of the coefficient for the 75^{th} quantile is higher than for the 50^{th} quantile, which is lower than the coefficient calculated on the 25^{th} quantile. In the RS, the Wald test shows that the hypothesis of equality among the two coefficients can be rejected at 1% confidence level. In the FBiH, the coefficients of the 25^{th} and 50^{th} quantile regression can be regarded as statistically equal. The coefficient of the 75^{th} quantile regression is instead statistically different than the one calculated on the 50^{th} quantile. In the RS, the presence of a household head who is a temporary resident negatively impacts on the household consumption and the coefficient for poorer households is higher than the one of the richer percentile. However, on the basis of the Wald test, the hypothesis of equality between the coefficients of the three regressions cannot be rejected. In the FBiH, the coefficient is statistically significant in both the 50^{th} and the 75^{th} quantile.

At the same time, the values of the coefficients are lower compared to the RS. This finding suggests a very different situation of temporary residents in the two Entities. The coefficients for the gender of household head are not significant. It cannot therefore be concluded that female headed households (FHH) are poorer than male headed households (MHH). Rather, in substantial accordance with the literature on gender and poverty in Bosnia and Herzegovina, being a FHH can turn out to be a source of concern only when combined with other factors²⁶.

Education. The only education variable that negatively affects household well-being relates to individuals who do not know how to read and write. The coefficient is significant at the 1% confidence level in both the FBiH and RS²⁷. In the RS, the coefficient of this variable for the 25th quantile regression equation is almost twice lower than the coefficient estimated for the 70th quantile. This suggests that illiteracy is a concern especially for poorer population strata. However, the two coefficients are not statistically different. Looking at the other education variables, in the RS the only statistically significant coefficients are those related to higher education and those for female adults with secondary school qualifications. In the RS, the presence of individuals having a higher education diploma has a bigger impact on welfare compared to the FBiH for all the regression equations.

<u>Employment.</u> In the RS, the presence of pensioners in the household positively contributes to welfare only when the 50th quantile estimates are taken into consideration. Instead, in the FBiH it is negatively significant (at 5% confidence level) for the 75th quantile. This can be due to the fact that in a poorer environment (such as the Republic Srpka), the income source coming from pension is highly relevant. On the other hand, in a relatively more-developed location (such as the FBiH), the presence of pensioners has no impact on household consumption and, when the richest quantile is considered, the presence of pensioners is negatively associated with level of household welfare²⁸.

In both the RS and the FBiH, the number of unemployed over household labour force negatively and significantly affects household welfare. This finding contrasts with the

²⁶ Rukanesen (2003) identifies the following traits as determinants of poverty for FHH: the heads are elderly of pensioners, they are headed by widows living alone, the household is made up of refugees or displaced persons, they are based in the RS.

²⁷ Somewhat surprisingly, adult illiteracy is not regarded as a key problem for Bosnia and Herzegovina (PRSP, 2002) despite that recent researches available on human development show that the country records a general backwardness in terms of adult literacy compared to the other Easter European and Stability Pact countries (UNDP, 2002).

²⁸ However, further research would be needed to explore the situation of elderly and pensioners in the country.

conclusion of the Country Poverty Profile that does not consider unemployment a primary cause of poverty. In the FBiH, the presence of unemployed is more relevant for the poorest strata of population. However, no statistically significant difference is found between the coefficients of the 25th and 50th equation, whereas the coefficient of the 50th and 75th model should be regarded as statistically different.

By looking at household members involved in income generating activities, in both the FBiH and the RS, the highest coefficient refers to enterprise ownership. The presence of public employees positively influences household consumption. In both Entities, the value of the coefficient in the bottom quantile model is higher compared to the other models. The coefficients are regarded as statistically different only in the RS when the coefficients for the 50th and the 75th model are considered. On the basis of the regression results, loss of employment among public sector employees and the increase in unemployment is expected to have a major negative impact on the poorest households. At the same time, since in the FBiH, the value of the coefficient of public sector employees is higher than in the RS, it is reasonable to suppose that a larger welfare loss will be experienced by the households hit by the shock. The regression results for the variable referring to informal dependent labour are of more complex interpretation. For the RS, the estimated value of the coefficient is significant at the 10% confidence level only in the 75th quantile regression and at 5% for the 50th quantile model In the FBiH, the variable is significant only for the highest quintile model. This result suggests that territorial issues have a key role in determining the importance of household livelihoods. For the poorest percentiles, it is reasonable to attribute the non-significance of the coefficient to the low skills endowment of individuals belonging to low population strata which imply scarce access to well-paid informal jobs. Whereas for richer households, the variable becomes a relevant livelihood source.

Location. The F-test proves that the location variables are jointly significant in both Entities at the 5% confidence level. In the RS, the analysis of the coefficients shows the high negative impact on household welfare associated to Visegrad, compared to the benchmark constituted by Banja Luka. At a lower level of severity, also Prijedor, Srpska Ilidza, Knedzevo and Canjce show negative significant coefficients. In the FBiH, the result for Zenica and Breza are particularly severe if compared to the reference municipality of Central Sarajevo. To explain the negative coefficients of these variables, the economic history of the Bosnian municipalities, their diverse resource and infrastructure endowment should be discussed. The economic history of Bosnia Herzegovina has been in fact remarkably dissimilar across the territory; during the conflict of the 1990s, Bosnian municipalities experienced a very uneven pattern of destruction, as well as different experiences and difficulties in recovering (Bisogno and Chong, 2002; Cuna and Alasia, 2003).

6. Vulnerability Analysis – Shock on Formal Public Sector

<u>The Shock</u>. The first shock involves the households having at least one individual employed in the formal public sector. This sector refers to public enterprises, institutions and organisations regardless of occupation characteristics and sector of activity. The sector involves approximately 47% of the households in the RS and 39% in the FBiH. The shock consists in the employment loss of one household member compensated by an increase in

unemployment. No other changes are hypothesised in household livelihoods. The workers involved in the formal public sector can be considered at risk of losing employment for two main reasons. The first regards the necessity for Bosnia and Herzegovina to undergo a radical reform in the public sector. This is one of the recommendations contained in the "Public Expenditures and Institutional Review", developed by the Bosnia and Herzegovina World Bank in 2002. The amount spent to pay public employees in Bosnia and Herzegovina "is almost three-times the average of Central and Eastern Europe transition economies and higher than the most of the more developed EU economies" (World Bank, 2002, p.35)²⁹. The second issue relates to privatisations. The risk of dismissal is particularly high for individuals employed in public enterprises that are planned to be privatised in next few years. This is because, despite the legal obligations, newly privatised enterprises might require a drastic cutback in employment in the search for economic and productive efficiency (Vodopivec, 2002; PRSP, 2002).

<u>Vulnerability Figures</u>. Table 2 shows the initial (before the shock) poverty incidence and the vulnerability to poverty headcount which gives the number of households estimated to fall below the poverty line because of the shock.

	Republic Srpska	Federation
Benchmark Poverty Incidence	22.92%	9.96%
Vulnerability to Poverty Headcount	7.15%	17.98%
Benchmark Poverty Gap	0.092	0.057
After Shock Poverty Gap	0.108	0.084
Benchmark Poverty Severity	0.028	0.013
After Shock Poverty Severity	0.034	0.021
Number of Observations	950	1122
Percentage of Households ²	47%	40%

- Table 2 -Formal Public Sector Households - Vulnerability and Poverty Figures¹

1 – All the figures refer to the group of households having at least one household member involved in public sector occupations.

2 – All the indexes are calculated using analytical weights.

The figures exclusively refer to the group of households hit by the shock, hence those having at least one member employed in the public sector. The poverty estimates for the group of households are in accordance with the overall country figures that show a higher poverty incidence in the RS than the FBiH. However, poverty incidence among the group hit by the shock is lower than the estimated Entity poverty incidence³⁰. Looking at the "vulnerability to poverty" headcount, in accordance with expectations, the FBiH registers a higher vulnerability to poverty. As commented above, this is a direct consequence of the

²⁹ The report is very careful in saying that the Bosnia public sector is overstaffed. On one side the report states that "the high public wages spending is more the result of relatively high wage and salaries levels in Bosnia and Herzegovina public sector vis-à-vis the rest of the economy" (ibid, p.26). On the other, it clarifies that, in drawing the above conclusions, the level of employment in the defence sector is not taken into account. Since the employment level of the defence sector in Bosnia and Herzegovina is very large, it is reasonable to argue that public sector employment must be reviewed.

³⁰ According to the estimated value of household consumption, in the RS, poverty incidence is 27% before and 31% after the shock; poverty gap 0.035 and 0.041; poverty severity 0.011 and 0.013. In the FBiH, the poverty incidence is 15% and 22%; poverty gap 0.026 and 0.038; poverty severity 0.006 and 0.009.

regression estimates, where the coefficients referring to unemployment and public sector employees have a higher value in the FBiH compared to the RS.

The analysis of the poverty gap and the squared poverty gap indexes shows that the RS has higher depth and severity of poverty both before and after the shock. However, the shock provokes a higher increase of these indicators in the FBiH. Poverty gap increases of 47% in the FBiH compared to 17% in the RS; squared poverty gap increases of 61% in the FBiH and of 21% in the RS. These results suggest that the two Entities present considerable differences in terms of welfare mobility provoked by the shock.

This finding is confirmed by the analysis of transition matrices. Unfortunately, the whole information contained in transition matrices cannot be summarised in a unique synthetic indicator of welfare mobility. If the number of households losing more than two welfare deciles is taken into consideration, it can be concluded that 37% of the RS households hit by the shock experience "severe" welfare downturns, amounting to more than two deciles.

- Table 3 -

	Formal Public Sector– Transition Matrices RS ¹													
		Simulated-After Shock Population Deciles												
		1	1 2 3 4 5 6 7 8 9 10 Total											
	1	100										100.00		
u	2	59.29	40.71									100.00		
oulatio	3	2.43	61.08	36.50								100.00		
	4		0.31	84.17	15.52							100.00		
op es	5		6.22	31.91	61.87							100.00		
k I šcil	6			5.34	47.70	24.74	22.21					100.00		
De	7			18.97	28.55	14.38	36.14	1.97				100.00		
hn	8			0.22	4.86	28.23	47.23	15.79	3.68			100.00		
enc	9					0.21	35.26	54.30	16.28	3.95		100.00		
Ä	10						0.96	17.15	27.13	32.06	22.70	100.00		
	Total	16.40	10.80	17.62	16.21	6.76	13.00	8.75	4.65	3.57	2.25			

1 - The transition matrix is calculated by taking into consideration the household population hit by the shock. The cut-off points are established on the basis of the estimated log of household consumption per capita and have the following values: 7.31, 7.47 7.64, 7.76, 7.83, 7.98, 8.10, 8.22, 8.41. Analytical weights are applied.

Regardless of this indicator, the analysis of Table 3 evidences that, in the RS, upper deciles register the highest shares of households losing more than two welfare deciles. For example, only 2% of households remain the 7th decile. Very similar is the case of the 8th and 9th decile. In many cases, considerable shares of households lose two or more deciles. Particularly severe is the downward welfare mobility experienced by households previously ranked in the 8th and 9th deciles, where more than 80% of the households experience welfare loss amounting to at least two deciles. Welfare losses decrease as one moves towards lower decile. In the 6th decile, the share of population losing more than two deciles is approximately 60% and decreases to marginal shares in the 3rd and 4th decile. This is a consequence of the regression coefficients but it also depends on the characteristics of the population involved in the formal public sector. These figures should not be confused with the overall poverty situation in the RS, which show high poverty gap and severity indexes. On this argument, it is important to notice that after the shock, approximately 45% of the population of households involved in the formal public sector is categorised in the bottom three deciles (which approximately corresponds to the poverty line).

Vulnerability to poverty especially involves households ranked immediately above the poverty line, namely those categorised in the 4th decile. Only 15% of these households remain on a non-poor status after the shock. Vulnerability to poverty is also experienced by groups of households from the 5th and 7th quantile, whereas it is rather limited from the 6th quantile.

The results for the FBiH are significantly different (table 4). Even the visual structure of the table suggests a higher downward welfare dynamism provoked by the shock. According to the estimates, 47% of the households hit by the shock experience welfare losses amounting to at least two deciles. Similarly to the RS, in the FBiH upper deciles register the highest downward welfare mobility. At the same time, it is important to notice that in all the deciles, very few households are capable of preserving their welfare ranking³¹. If the population classified in the upper part of the distribution is considered, in the FBiH, approximately 28% of the population is classified in the top group. This share increases to 32% in the RS.

Table 4 shows that the 8th and 9th deciles are the ones recording the highest number of households losing more two welfare deciles after the shock (72% and 83%). This share decreases to 69% in the 7th and decreases until 44% in the 5th decile. However, contrarily to what evidenced in the RS, also lower deciles record considerable number of households vulnerable to severe welfare losses. This is the case of 29% of households from 4th and 61% from the 3rd decile losing more than two deciles. At the same time, it is important to notice that the movement of households below the poverty line in the FBiH is higher compared to the RS: all the households previously categorised in the 2nd decile fall to the bottom decile. This explains the higher increase in poverty gap in the FBiH compared to the RS summarised in Table 2.

		Simulated-After Shock Population Deciles FED												
		1	2	3	4	5	6	7	8	9	10	Total		
	1	100.										100.0		
_	2	100	0.00									100.0		
pulation	3	61.49	35.3	3.14								100.0		
	4	11.20	18.0	70.8	0.00							100.0		
	5	1.82	3.61	38.8	52.5	3.26						100.0		
\mathbf{P}_{0}	6	4.22	0.00	7.49	43.3	45.0	0.00					100.0		
rk	7		5.05	5.50	25.8	33.1	29.2	1.35				100.0		
ma	8			0.56	6.93	29.2	35.9	24.7	2.74			100.0		
chı iles	9				3.34	5.64	28.5	45.6	15.5	1.44		100.0		
en	10				1.20	0.00	0.35	19.7	36.9	34.0	7.77	100.0		
ВЦ	Total	27.94	6.22	12.7	13.3	11.6	9.43	9.03	5.43	3.52	0.77			

- Table 4 -Formal Public Sector – Transition Matrices FBiH¹

1 - The transition matrix is calculated by taking into consideration the population of households hit by the shock. The cut-off points are established on the basis of the estimated log of household consumption per capita and have the following values: 7.51,7.60, 7.72, 7.84, 7.96, 8.06, 8.21, 8.39, 8.59. Analytical weights are applied.

Vulnerability to poverty mainly involves households from the 3rd and 4th decile. In particular, 93% of the households living immediately above the poverty line (3rd decile) fall into poverty after the shock. In the RS, "only" 85% of the households immediately above the

³¹ This information is given by looking at the diagonal in the matrices in which the population decile before the shock is equal to the one after the shock.

poverty line are estimated to be plunged in poverty³². These figures help to explain the different "vulnerability headcount" presented in Table 2.

Although it might sound anomalous, a possible explanation of the higher welfare mobility of the FBiH can be found in the higher living standards of this Entity compared to the RS. An environment characterised by higher industrial and economic growth, better infrastructure endowment, etc., encourages households to rely on their income sources. This could be one of the reasons for the higher coefficients associated with the presence of public sector workers in the household. Consequently, when the income-generating activity is subtracted, households are more exposed to welfare losses. In contrast, the worse economic condition of the RS forces households to rely on multiple or alternative livelihood sources such as un-registered occupations, consumption of goods obtained from direct agricultural activity, remittances from relatives living abroad, etc. Following this rationale, disappearance of income source has a lower impact on welfare mobility³³.

Another interpretation relates to the distribution of employees across the formal public sector. In the RS, 17% of the individuals employed in the public sector perform "simple occupations"³⁴. The share decreases to 13% in the FBiH³⁵. For the RS, the major concentration of public sector workers in simple occupations, and consequently their lower wages, decreases the average return to this type of employment. The removal of such low-wages income sources does not create relevant welfare loss as is the case of more remunerative jobs. This interpretation must be treated very carefully. Saying that households involved in the public sector in the RS are less vulnerable to welfare losses than those in the FBiH because the former are more concentrated in low-wage activities sounds very unrealistic. Workers engaged in simple occupations may in fact be considered at a higher risk of dismissal compared to other typologies of workers (Sinha and Lipton, 1999). If this hypothesis holds true, although the RS records overall lower welfare mobility, it is reasonable to suppose that the risk of employment loss among households involved in public sector occupations is higher in the RS.

Despite the higher mobility verified in the FBiH, the after-shock Entity poverty headcount in the RS is still higher than in the FBiH (31% vs. 22%). It must be considered that the benchmark poverty incidence in the RS is almost twice that of the FBiH. However, the picture in terms of vulnerability to poverty level would significantly change if, in order to draw the poverty line, a relative approach is adopted on the basis of the Entity-specific welfare distribution. By utilising a relative poverty line corresponding to the cut-off point of the second decile with respect to the before-the- shock distribution, the poverty incidence after the shock would be higher in the FBiH than the in the RS (26% vs. 23%)³⁶.

<u>Vulnerability Profile</u>. The shock pushes some households into poverty; others are hit by the shock but were already in a poverty status, others remain on a non-poor welfare level

³⁵ Detailed results available under request.

 $^{^{32}}$ In the transition matrix elaborated for the FBiH, the poverty line can be approximated to the cut-off point of the 2nd to the 3rd decile.

 $^{^{33}}$ For the FBiH, the estimated coefficient is 0.197 versus 0.140 in the RS. Similarly the presence of unemployment is not statistically relevant in the RS whereas it is significantly negative in the FBiH.

³⁴ Simple occupations include the following categories: workers in services and commerce; agriculture, forestry and fishing; mining, civil engineering, processing and transport workers; other undefined simple occupations.

³⁶ Needless-to-say, if this "relative" definition of poverty is adopted, the benchmark poverty headcount is 20% in both Entities, since 10% of the household population live in each decile.

but experience downward movements in the welfare ranking. The vulnerability profile in table 5 evidences the household characteristics that are associated with vulnerability to poverty and to what extent they differ from households who are hit by the shock but were already in a poverty status and from those who, although hit by the shock, remain on a non-poor status.

By first looking at the results of the RS, households whose heads are illiterate or have no diploma are more at risk of being plunged into poverty. They are also more likely to be hit by the shock when they are already poor. The percentage of households who fall into poverty because of the shock decreases when education of the household head increases. In the FBiH on the other hand, illiterate or no-diploma headed households are not the category at higher risk of poverty. Rather, vulnerability to poverty is experienced by primary or secondary school qualified headed households.

]	Poverty	Dynami	cs			Welfare	Dynamics	3	Share of	
Household Characteristics		Republic Srpska		Bosni	Federation of Bosnia and Herzegovina			ublic oska	Federa Bosni Herze	ntion of a and govina	Population	
	Poor	Vuln. To Pov.	Non Poor	Poor	Vuln. To Pov.	Non Poor	< 2 Dec.	> 2 Dec.	< 2 Dec.	> 2 Dec.	RS	Fed.
MHH	24.4	7.8	67.7	10.6	19.4	69.2	71.4	28.6	55.4	44.6	90.0	88.3
FHH	9.3	8.3	82.6	5.2	7.5	87.2	55.7	44.3	49.0	51.0	9.9	11.7
Ill. /No Dip.	31.3	18.4	50.4	15.2	11.6	73.2	81.6	18.4	69.3	30.7	11.6	9.7
Prim. Dip.	29.6	4.7	65.7	15.8	20.3	63.9	74.1	25.9	75.2	24.8	22.5	15.3
Second. Dip.	18.3	7.4	74.3	9.5	21.1	69.4	65.1	34.9	49.7	50.3	53.7	59.6
Higher. Dip.	10.6	6.7	82.6	1.0	7.4	91.6	62.6	37.4	42.6	57.3	12.2	15.4
<=2 Mem.	0.00	1.7	98.3	0.0	0.0	100.0	53.6	46.4	40.6	59.4	6.16	6.9
3 Mem.	6.0	5.4	88.6	0.6	5.3	94.1	56.6	43.4	37.0	63.0	17.6	18.8
4 Mem.	16.1	11.6	72.2	5.0	20.7	74.2	73.6	26.4	40.3	59.7	35.2	39.6
5 Mem.	24.7	5.0	70.2	18.1	32.8	49.1	66.4	33.6	73.7	26.3	22.7	20.2
6+ Mem.	57.5	8.5	34.0	29.0	14.7	56.3	84.8	15.2	97.1	2.8	18.4	14.5

- Table 5 -Formal Public Sector– Poverty and Vulnerability Profile

In the FBiH, MHH have the highest incidence of vulnerability to poverty. This confirms the finding that being a FHH is not *per-se* a cause of poverty. In the RS, the percentage of FHH that fall into poverty after the shock is slightly higher compared to MHH; however, FHH are less likely to be found already in poverty when the shock occurs. Looking at the results for household size, the results show that in both Entities, when the shock involves large households, a significant share of them is already in a poverty status. This confirms the concern of the Poverty Profile for the condition of children living in large households in the country. This is particularly the case for RS, where 57% of the households having 6 or more members, are already in a poverty status before the shock. On the other hand, vulnerability to poverty is almost non-existent among single or two-member households.

When vulnerability is analysed by considering the welfare losses amounting to more than two deciles, the scenario drastically changes. FHH are the category more exposed to severe welfare losses: in the RS, out of 100 FHH hit by the shock 44% of them experiences welfare losses amounting to more than two deciles. This share increases to 51% in the FBiH. This result contrasts with the conclusion of the "vulnerability to poverty" approach. Welfare losses especially involve households whose head has a diploma: this is due to the fact that in educated-headed households, loss of employment is likely to correspond to the loss of a

relatively higher wage. Contrarily to that previously discussed welfare losses are not influenced by household size: rather, in both Entities, households made up of few members register the highest share of households losing ranking positions. This can be due to the scarce diversification of the livelihood sources for small-size households. Hence, the subtraction of an income source provokes welfare losses regardless of poverty considerations.

7. Vulnerability Analysis – Shock on Informal Dependent Sector

The Shock. This shock involves the households whose members are employed in the informal dependent sector. According to LSMS estimates, 17.5% of the total households living in Bosnia and Herzegovina have at least one individual engaged in an informal dependent occupation (19.2% in the RS and 14.7% in the FBiH). Similar to the shock analysed above, this second shock is modelled by hypothesising that in all the households involved in the informal dependent sector, one member loses his/her work and becomes unemployed. As specified above, this category encompasses all the employees in private enterprise who do not receive pension contributions. This category is exposed to peculiar typologies of risk and insecurity. Employees hired without fully respecting the registration procedures enjoy less protection or labour rights and the risk of dismissal is very high. This is confirmed by the individual data from the LSMS which show the low duration of informal employment compared to others. Informal dependent employees are also exposed to the consequence of policies aimed at regulating informal labour which might provoke contractions in the labour demand and consequent loss of employment. These issues emerge in various researches conducted in Bosnia and Herzegovina (Djipa et al., 2002, UNDP, 2002).

<u>Vulnerability Figures</u>. According to the estimates, in the RS 40% of the households engaged in this sector is already poor before the shock. In the FBiH, the incidence is considerably lower (12.89%). However, this figure is higher compared to the group of households involved in formal public occupations.

- Table 6 - Informal Dependent Sector - Vulnerability Headcount ¹								
	Republic Srpska	Federation						
Benchmark Poverty Incidence	40.32%	12.89%						
Vulnerability to Poverty Headcount	4.32%	7.26%						
Benchmark Poverty Gap	0.236	0.245						
After Shock Poverty Gap	0.252	0.264						
Benchmark Poverty Severity	0.073	0.059						
After Shock Poverty Severity	0.079	0.065						
Number of Observations	403	282						
Percentage of Households ²	19.2%	14 7%						

1 - All the figures refer to the group of households having at least one household

member involved in public sector occupations.

2 – Percentage of households is calculated using analytical weight from the LSMS.

For the households involved in the informal dependent sector, the FBiH has deeper poverty but lower severity compared to the RS. The households living in the FBiH experience higher vulnerability to poverty compared to their analogous households in the RS (7.6% vs. 4.3%). It has to be noticed that the vulnerability to poverty figures associated to this

shock are lower compared to the case of a shock occurring in the formal public sector. Poverty gap increases of approximately 7% in the two Entities (6.78% in the RS and 7.76 in the FBiH). The increase in poverty severity is instead considerably higher in the FBiH than in the RS (10.17% vs. 8.22%). However, the RS has still the highest poverty severity index among the households hit by the shock.

A shock involving informal dependent labourers has also a relatively lower impact in terms of welfare dynamics. From transition matrices, it is noticeable that in the RS, the categories experiencing the most severe welfare losses belong to high deciles. In the 10th decile, the population losing more than two deciles after the shock is approximately 49% and increases to 59% in the 9th decile. Approximately 13% of the total population living in the top three deciles before the shock remain on that level after the simulation. The percentage of households losing more than two deciles decreases if lower deciles are taken into consideration. By considering the entire population hit by the shock, the ratio of households experiencing a welfare loss amounting to two deciles or more is 14.6%, which is considerably lower with respect to the estimates made for formal public sector employees.

In the transition matrix elaborated in Table 7, the poverty line can be drawn approximately between the 4^{th} and the 5^{th} decile. Vulnerability to poverty involves slightly less than 40% of the households categorised in the 5^{th} decile before the shock. For other deciles, the percentage of households falling into poverty is marginal. At the same time, it is important to notice that also households living below the poverty line experience downward welfare mobility. This means that for more than 40% of the households already in poverty before the occurrence of the shock, loss of employment implies more severe poverty and destitution.

		Simulated-After Shock Population Deciles												
		1	2	3	4	5	6	7	8	9	10	Total		
	1	100.00										100.00		
oulation	2	31.19	68.81									100.00		
	3		53.71	46.29								100.00		
	4			26.50	73.50							100.00		
Pop	5			0.00	39.49	60.51						100.00		
lk l ecil	6			4.05	0.00	40.47	55.48					100.00		
Do	7				0.75	11.51	47.16	40.57				100.00		
hn	8						40.83	53.38	5.79			100.00		
enc	9							59.15	29.87	3.42		100.00		
Ā	10							2.71	27.43	33.43	36.42	100.00		
	Total	14.84	11.18	7.34	11.29	11.32	14.92	15.67	6.26	3.62	3.57			

- Table 7 -	
Informal Dependent Sector– Transition Matrices	RS

1 - The transition matrix is calculated by taking into consideration the population of households hit by the shock. The cut-off points are established on the basis of the estimated log of household consumption per capita and have the following values: 6.93, 7.21, 7.33, 7.51, 7.67, 7.87, 8.05, 8.16, and 8.41. Analytical weights have been applied.

The transition matrix elaborated for the FBiH confirms the overall higher vulnerability to welfare losses of households living in this Entity. However, the welfare losses are not as dramatic as in the case of formal public employees. The ratio of households losing more than two welfare deciles out of the total population hit by the shock is 24% which is almost half the value estimated for shocks occurring in the formal public sector. Examining the indicators of welfare mobility for upper deciles, according to the simulation

results, 14% of the households previously categorised in the top three deciles are still on the same welfare level after the shock. The transition matrix of the FBiH shows that the shock especially involves households categorised in the 6th decile. Only 3% of the households classified in this decile are capable of preserving their living standards after the loss of the livelihood source associated to informal dependent occupations.

Vulnerability to poverty is particularly relevant for households classified in the deciles immediately above the poverty line: almost all the households previously in the 3rd decile fall in poverty after the shock³⁷. Poverty is also experienced by 15.5% of households from the 4th and by a 6.8% from the 5th decile. These figures are coherent with the vulnerability to poverty headcount presented in Table 6 which shows that in the FBiH a bigger number of households are vulnerable to poverty compared to the RS. Again, vulnerability to poverty is a consequence of the higher exposure to downward welfare movements registered in the FBiH. As commented above in note to regression results, the overall better economic condition of informal employees living in the FBiH should be taken into consideration. At the same time, the wage figures show that the average wage of informal dependent employees in the RS is KM 292 per month whereas in the FBiH it is KM 464 per month. As already mentioned, when a well-paid income source is subtracted, welfare losses tend to be higher. This is the case of households involved in the informal dependent sector in the FBiH. On the other hand, the low remuneration of informal dependent employees is associated to poverty, as confirmed by the RS poverty figures in Table 6 but decreases the household vulnerability to welfare downturns.

		Simulated-After Shock Population Deciles FED												
		1	2	3	4	5	6	7	8	9	10	Total		
	1	100.0										100.0		
pulation	2	69.00	31.00									100.0		
	3	17.19	80.52	2.29								100.0		
	4		15.50	54.37	30.13							100.0		
	5		6.85	48.66	15.15	29.3						100.0		
\mathbf{P}_{0}	6				0.81	49.2	50.03					100.0		
rk	7				5.77	18.1	72.99	3.15				100.0		
ma	8				1.56	13.1	11.64	60.39	13.30			100.0		
ch1 iles	9					1.53	9.12	49.11	27.53	12.71		100.0		
en	10							8.48	40.40	39.10	12.02	100.0		
рв	Total	18.76	13.48	10.52	5.39	11.0	14.45	12.23	7.99	5.02	1.15			

- Table 8 -Informal Dependent Sector– Transition Matrices FBiH

1 - The transition matrix is calculated by taking into consideration the population of households hit by the shock. The cut-off points are established on the basis of the estimated log of household consumption per capita and have the following values: 7.49, 7.64, 7.72, 7.78, 7.90, 8.01, 8.17, 8.32, and 8.50. Analytical weights have been applied.

<u>Vulnerability Profiles</u>. In the RS, the analysis of the effects of the shock amongst the various population categories evidences that female and illiterate headed, together with large households, are already in a poor status before the shock. Looking at gender characteristics, out of 100 female headed households hit by the shock 50 of them are already in a poverty status before the shock. This share increases to 60% for households headed by someone having no diploma or incapable to read and write. At the same time, 77% of the households having more than 6 members are instead in poverty before the shock occurrence. These

³⁷ With a certain degree of approximation, the cut-off point of the second and the third decile can be considered as the poverty line.

categories should be regarded as highly in need of social protection due to the high insecurity associated to informal dependent labour.

The profile for households vulnerable to poverty presents various dissimilarities with respect to the one of the "always poor". First of all, female headed households involved in informal dependent occupations seem not to experience any risk of being plunged into poverty. It must be said that almost the majority of them are already in a poor status, but the remaining families are kept on a non-poor welfare status after the disappearance of the informal income source. In contrast, male headed households are vulnerable to poverty but the percentage is rather small since the Entity poverty figures are small too. In terms of education characteristics (proxied by the household head education attainment), the households more at risk of being plunged into poverty are those whose head has no diploma or are illiterate. The incidence of vulnerability to poverty among household-size categories suggests that large households involved in informal dependent occupation are in general already poor before the shock occurrence. Vulnerability to poverty is higher among three-member households but no specific reason can be found for explaining this result.

In the FBiH, the profile of households vulnerable to poverty in terms of gender of household head is very similar to the RS. Male head households are the category more likely to be plunged in poverty after the shock. Differently from the RS, FHH are not found among the poor population before the shock. At the same time, the simulation shows that the FHH involved in informal dependent occupations count on a quite secure livelihood system that allows them to remain on a non-poor status after the shock. The households headed by a well-educated head present the highest vulnerability to poverty whereas almost 25% of the illiterate/no diploma headed households are hit by the shock when already in a poverty status. The figures of household size show that small-size households are non-poor and not-vulnerable. Incidence of poverty and vulnerability increase when bigger size households are taken into consideration: it must be noticed that the category recording the highest vulnerability to poverty is the 4-members households.

			Poverty	Dynami	cs			Welfare	Dynamics	5	Share of		
Household		Republic			FBiH			RS		FBiH		Population	
Characteristics		Srpska Vuln			Vuln								
	Poor	to Poverty	Non Poor	Poor	to Poverty	Non Poor	< 2 Dec.	> 2 Dec.	< 2 Dec.	> 2 Dec.	RS	Fed.	
MHH	38.8	4.9	56.2	14.2	8.0	77.8	77.9	22.1	79.5	20.5	87.6	89.9	
FHH	50.8	0.0	49.2	1.3	0.8	97.9	83.0	17.0	84.1	15.8	12.4	10.1	
Illiterate/ No Dip.	60.3	8.4	31.3	23.3	4.4	72.2	80.7	19.3	83.4	16.6	25.3	27.6	
Prim. Dip.	38.6	0.8	60.6	11.6	12.2	76.2	81.1	18.9	84.9	15.1	32.5	20.1	
Second. Dip.	20.9	5.1	74.0	8.5	4.8	86.7	74.9	25.1	76.4	23.6	36.2	49.9	
High Edu. Dip.	24.7	6.9	68.4	0.0	35.0	65.0	53.6	46.3	65.4	34.6	6.0	2.4	
<=2 Mem.	9.2	1.8	88.9	0.0	0.0	100.0	54.7	45.3	49.21	50.8	5.5	3.4	
3 Mem.	11.5	10.8	77.7	1.0	1.5	97.5	58.6	41.4	49.0	51.0	15.6	15.8	
4 Mem.	10.8	0.6	88.7	5.9	11.2	82.8	69.8	30.2	79.5	20.5	30.2	27.3	
5 Mem.	60.2	6.3	33.4	14.2	5.8	80.0	88.5	11.5	78.5	21.5	16.3	18.6	
6+ Mem.	77.1	4.1	18.7	24.6	8.4	67.0	95.3	4.7	98.8	1.2	32.3	34.5	

- Table 9 -Informal Dependent Sector – Poverty and Vulnerability Profile

The profile of welfare dynamics shows that FHH are not exposed to welfare losses as they are in the case of shock occurring in the formal public sector. This holds both in the RS and in the FBiH. Considering the qualifications of the household head, well-educated headed households are the category more exposed to severe welfare downturns. It is possible to argue that this is related to the higher remuneration of informal employment for well-educated households. It should be specified that the shock does not hypothesise which member of the household will lose the job. However, it is reasonable to suppose that well-educated headed households can have access to better-paid occupations. Finally, as in the previous shock model, welfare losses are negatively related to household size: in both Entities, households composed by few members register the highest share of households losing ranking positions. It can be supposed that small-size households have scarcer possibilities of income diversification and therefore the deletion of an income source provokes welfare losses.

8. Conclusive Remarks

In this paper, a simulation methodology was adopted to assess at what extent selected employment shocks are likely to create poverty and welfare downturns in Bosnia and Herzegovina households. According to the findings of this paper, the outcomes of a shock in the formal public sector are considerably diverse with respect to the case of a shock in the informal dependent sector. At the same time, the characteristics of the households vulnerable to poverty confirm that already stated by numerous researches, such as Baulch and McCulloch (1998); Prichett et al. (2000): the profile of the vulnerable does not necessarily coincide with one of poverty. Also, the definition of vulnerability is important for detecting which categories are in need of social protection.

On a methodological point of view, the adoption of a quintile regression has allowed us to overcome one of the most recurrent limitations of micro-simulation literature: a variable can have diverse effect on household welfare depending on the relative position of the household in the welfare distribution. Further research would be needed to integrate microsimulation with probability statement regarding the occurrence of a negative event. Also, the availability of panel data can help evaluate the reliability of the estimates made on the basis of micro-simulations.

Descriptive s		10	10.05			
Variable Name	All	<10	10-25	25-50	50-75	75-100
	Households	%Tile	%Tile	%Tile	%Tile	%Tile
Natural Logarithm of Household Consumption	7.960	7.091	7.572	7.845	8.090	8.533
Natural Logarithini of Household Consumption	(0.520)	(0.301)	(0.081)	(0.072)	(0.073)	(0.275)
Demons 0. 7 weeks ald	0.274	0.495	0.460	0.267	0.197	0.137
Persons 0- / years old	(0.607)	(0.835)	(0.756)	(0.555)	(0.516)	(0.406)
D 0.14 11	0.257	0.386	0.296	0.327	0.243	0.141
Persons 8 -14 years old	(0.533)	(0.679)	(0.602)	(0.621)	(0.525)	(0.389)
	1.013	1.185	1.021	1.089	1.058	0.854
Females 15-59 years old	(0.855)	(0.943)	(0.864)	(0.886)	(0.842)	(0.768)
	0.993	1.278	1.050	0.977	1.018	0.832
Males 15-59 years old	(0.900)	(0.979)	(0.903)	(0.877)	(0.920)	(0.829)
	1.036	1 312	1 173	1 1 1 4	1 043	0.788
Log of household size	(0.556)	(0.531)	(0.530)	(0.497)	(0.503)	(0.553)
	0 243	0.357	0.320	0.248	0.207	0.171
Dummy, 1 if hh head is temp. resident	(0.243)	(0.470)	(0.320)	(0.432)	(0.406)	(0.171)
	0.428)	(0.479)	(0.407)	0.106	(0.400)	0.377)
Dummy, 1 if hh head is a women	(0.210)	(0.146)	(0.208)	0.190	(0.424)	0.238
	(0.413)	(0.330)	(0.407)	(0.397)	(0.424)	(0.438)
Age of household head	55.004	30.309 (12.757)	35.004	55.150	54.858	54.008
•	(13.852)	(13.757)	(14.685)	(13.870)	(13.117)	(13.857)
Square of age of household head	3217.31	338.817	3313.59	3232.05	3181.191	3108.65
	(1531.49)	(1552.25)	(1660.6)	(1528.73)	(1461.27)	(1493.21)
Adults unable to read/write on hh size	0.067	0.128	0.084	0.069	0.056	0.036
really and to read write on hir size	(0.209)	(0.269)	(0.230)	(0.215)	(0.192)	(0.163)
Ad with no qualifications on hh size	0.255	0.289	0.275	0.263	0.260	0.221
Ad. with no quantications on his size	(0.363)	(0.322)	(0.363)	(0.365)	(0.368)	(0.375)
Mala ad with primary school on hh size	0.086	0.115	0.090	0.093	0.078	0.072
Male ad. with primary school. on his size	(0.169)	(0.175)	(0.159)	(0.169)	(0.160)	(0.176)
	0.105	0.110	0.104	0.112	0.095	0.105
Fem. ad. with primary school. on hh size	(0.187)	(0.150)	(0.166)	(0.173)	(0.173)	(0.227)
	0.189	0.166	0.187	0.173	0.201	0.202
Male ad. with secondary school on hh size	(0.227)	(0.198)	(0.217)	(0.200)	(0.216)	(0.264)
	0.160	0.102	0.136	0.160	0.178	0.187
Fem. ad. with secondary school on hh size	(0.219)	(0.148)	(0.186)	(0.199)	(0.225)	(0.262)
	0.046	0.014	0.023	0.034	0.047	0.081
Male adults with higher education on hh size	(0.131)	(0.059)	(0.023)	(0.101)	(0.116)	(0.185)
	0.026	0.007	0.011	0.015	0.029	0.049
Fem. adults with higher education on hh size	(0.102)	(0.046)	(0.057)	(0.067)	(0.02)	(0.147)
	0.189	0.131	0.167	0.169	0.197	0.233
N° of pensioners on hh size	(0.227)	(0.222)	(0.267)	(0.287)	(0.222)	(0.036)
	(0.227)	(0.223)	(0.207)	(0.287)	(0.322)	(0.030)
N° of unemployed on hh adults	0.141	0.234	(0.180)	0.139	(0.122)	0.080
	(0.236)	(0.207)	(0.231)	(0.248)	(0.207)	(0.202)
N° of entrepreneurs on hh adults	0.026	0.008	0.010	0.013	0.030	0.049
1	(0.102)	(0.057)	(0.0/1)	(0.077)	(0.106)	(0.151)
N° of inf. dep. employees on hh adults	0.061	0.068	0.068	0.064	0.058	0.055
1 1 2	(0.165)	(0.149)	(0.1/1)	(0.172)	(0.153)	(0.170)
N° of infgen employees on hh adults	0.055	0.053	0.041	0.059	0.057	0.060
i or init geni emproyees en ini uuuno	(0.192)	(0.183)	(0.158)	(0.188)	(0.197)	(0.210)
N° of formal public employees on hh ad	0.188	0.126	0.168	0.184	0.203	0.219
iv of formal public employees on milda.	(0.279)	(0.219)	(0.255)	(0.273)	(0.270)	(0.317)
N° of formal private employees on hh ad	0.031	0.012	0.028	0.025	0.035	0.041
iv of format private employees on infad.	(0.110)	(0.063)	(0.102)	(0.098)	(0.117)	(0.131)
Size of the land used	0.401	0.420	0.284	0.445	0.309	0.432
Size of the fand used	(1.251)	(1.006)	(0.756)	(1.590)	(0.987)	(1.439)
Dummy, 1 if living in Banja Luka	0.390	0.235	0.251	0.378	0.438	0.504
Dummy, 1 if living in Srpska Ilidza	0.035	0.012	0.066	0.068	0.022	0.014
Dummy, 1 if living in Cajnice	0.015	0.018	0.018	0.008	0.024	0.009
Dummy, 1 if living in Novi Grad (RS)	0.065	0.061	0.066	0.078	0.068	0.054
Dummy, 1 if living in Prijedor	0.180	0.371	0.269	0.167	0.126	0.092
Dummy 1 if living in Modrica	0.055	0.034	0.039	0.053	0.064	0.066
Dummy 1 if living in Vicegrad	0.035	0.089	0.079	0.026	0.018	0.004
Dummy, 1 if living in Knodzevo	0.025	0.059	0.079	0.020	0.010	0.004
Dummy 1 if living in Samaa	0.025	0.038	0.029	0.020	0.020	0.015
Dummy, 1 if living in Zvornik	0.045	0.021	0.034	0.055	0.050	0.039
Dunniny, 1 if living in Schoo	0.105	0.049	0.121	0.118	0.105	0.115
Dummy, 1 if fiving in Stoac	0.030	0.040	0.023	0.045	0.034	0.000

 Table §.1

 Descriptive Statistics of Regression Variables – RS

Descriptive Statistics of Regi		labics –	r biii oi	Dosina ai	iu nei zeg	ovina
Variable Name	All Housebolds	<10 %Tile	10-25 %Tile	25-50 % Tile	50-75 %Tile	75-100 %Tile
	8 068	7 308	7 729	7.966	8 213	8 701
Natural Logarithm of Household Consumption	(0.520)	(0.277)	(0.070)	(0.066)	(0.082)	(0.208)
	(0.330)	(0.277)	(0.070)	(0.000)	(0.082)	(0.308)
Persons 0- 7 years old	(0.583)	(0.300)	(0.592)	(0.596)	(0.193)	(0.260)
	0.283	0.548	0.366	0.321	0.100	0.103
Persons 8 -14 years old	(0.578)	(0.747)	(0.642)	(0.521)	(0.483)	(0.356)
	0.989	1 227	(0.042)	1.061	0.951	0.734
Females 15-59 years old	(0.828)	(0.849)	(0.867)	(0.815)	(0.801)	(0.745)
	0.939	1 188	1 1 1 1	0.978	0.891	0.694
Males 15-59 years old	(0.860)	(0.858)	(0.860)	(0.860)	(0.846)	(0.803)
	0.988	1 306	1 195	1 079	0.933	0.649
Log of household size	(0.547)	(0.433)	(0.455)	(0.475)	(0.489)	(0.551)
	0.123	0.212	0.127	0.122	0.097	0.084
Dummy, 1 if hh head is temp. resident	(0.329)	(0.409)	(0.333)	(0.328)	(0.296)	(0.278)
	0.273	0.162	0 201	0.259	0.283	0.385
Dummy, 1 if hh head is a women	(0.445)	(0.369)	(0.401)	(0.438)	(0.450)	(0.487)
	53 957	49 925	53 114	53 916	55 2.60	56 069
Age of household head	$(14\ 171)$	(14555)	(14.038)	$(14\ 272)$	(13599)	(13794)
	3112.13	2704 01	3017 75	3110.38	3238 37	3333.83
Square of age of household head	(1551-75)	(1561.42)	(1531.17)	(1586.63)	(1517.75)	(1506.26)
	0.050	0.068	0.054	0.052	0.052	0.032
Adults incapable to read/write on hh size	(0.185)	(0.201)	(0.187)	(0.181)	(0.192)	(0.168)
	0.196	0.216	0.206	0.201	0.196	0.175
Ad. with no diploma on hh size	(0.320)	(0.296)	(0.301)	(0.309)	(0.326)	(0.348)
	0.073	0.113	0.089	0.080	0.064	0.042
Male ad. with primary school on hh size	(0.163)	(0.171)	(0.158)	(0.169)	(0.164)	(0.147)
	0.113	0.133	0.119	0.119	0.111	0.096
Fem. ad. with primary school on hh size	(0.202)	(0.170)	(0.170)	(0.191)	(0.202)	(0.241)
	0.212	0.172	0.221	0.233	0.232	0.203
Male ad. with secondary school on hh size	(0.239)	(0.178)	(0.208)	(0.219)	(0.243)	(0.291)
	0.174	0.105	0.143	0.166	0.196	0.225
Fem. ad. with secondary school on hh size	(0.244)	(0.161)	(0.174)	(0.203)	(0.248)	(0.318)
	0.055	0.015	0.029	0.034	0.061	0.105
Male adults with high diploma on hh size	(0.153)	(0.067)	(0.096)	(0.112)	(0.146)	(0.221)
	0.037	0.004	0.011	0.019	0.032	0.089
Fem. adults with high diploma on hh size	(0.139)	(0.033)	(0.059)	(0.086)	(0.116)	(0.221)
	0.248	0.142	0.199	0.217	0.286	0.338
N° of pensioners on hn size	(0.352)	(0.235)	(0.290)	(0.306)	(0.375)	(0.425)
No of uncomplexied on hh adulta	0.147	0.242	0.186	0.147	0.127	0.080
N° of unemployed on nn adults	(0.247)	(0.287)	(0.246)	(0.230)	(0.234)	(0.213)
No of ontrongeneous on hh adulta	0.016	0.004	0.006	0.011	0.023	0.027
N of entrepreneurs on hir adults	(0.088)	(0.045)	(0.048)	(0.068)	(0.100)	(0.120)
No aftinf dan amplayaas on hh adults	0.034	0.037	0.034	0.039	0.039	0.027
N of hir. dep. employees on hir adults	(0.125)	(0.119)	(0.117)	(0.126)	(0.134)	(0.125)
No aftinf can amplevees on hhadults	0.024	0.036	0.031	0.026	0.016	0.017
N of hir. gen. employees on hir adults	(0.116)	(0.140)	(0.128)	(0.114)	(0.098)	(0.105)
N° of formal nublic amployage on bh adults	0.176	0.135	0.151	0.187	0.181	0.206
N of formal public employees of fin adults	(0.266)	(0.213)	(0.223)	(0.260)	(0.268)	(0.313)
N° of formal private employees on hh adults	0.042	0.024	0.042	0.037	0.046	0.056
IN of formal private employees on fin addits	(0.139)	(0.092)	(0.141)	(0.122)	(0.137)	(0.171)
Size of the land used	0.298	0.860	0.340	0.261	0.086	0.098
Size of the land used	(5.979)	(12.84)	(4.696)	(3.881)	(0.546)	(0.570)
Dummy, 1 if living in Centar	0.087	0.022	0.041	0.070	0.082	0.170
Dummy, 1 if living in Novigrad (Fed)	0.147	0.110	0.114	0.157	0.168	0.169
Dummy, 1 if living in Nsarajevo	0.091	0.024	0.067	0.077	0.100	0.151
Dummy, 1 if living in Zenica	0.156	0.304	0.234	0.162	0.101	0.054
Dummy, 1 if living in Tuzla	0.175	0.186	0.162	0.171	0.199	0.162
Dummy, 1 if living in Vogosca	0.027	0.020	0.021	0.041	0.029	0.026
Dummy, 1 if living in Travnik	0.063	0.057	0.070	0.068	0.067	0.058
Dummy, 1 if living in Visoko	0.051	0.080	0.050	0.057	0.055	0.028
Dummy, 1 if living in Breza	0.019	0.016	0.046	0.028	0.016	0.004
Dummy, 1 if living in Zavidovici	0.048	0.046	0.065	0.052	0.043	0.0397.
Dummy, 1 if living in Gradacac	0.047	0.048	0.037	0.030	0.055	0.059
Dummy, 1 if living in Posusje	0.015	0.005	0.024	0.014	0.016	0.019
Dummy, 1 if living in Kakanj	0.047	0.069	0.050	0.054	0.046	0.029
Dummy, 1 if living in Grude	0.015	0.007	0.013	0.012	0.017	0.023

 Table §.2

 Descriptive Statistics of Regression Variables – FBiH of Bosnia and Herzegovina

Bibliography

Alasia, A., Soregaroli, C., (2002). "Economic Growth and Food Demand Changes in Mozambique: an Assessment of Scenarios and Implication for Policy", Cid Working Paper N° 11. Bologna.

Amin S., Ray, A.S., Topa G., (2001). "Does Microcredit Reach the Poor and Vulnerable? Evidence from Northern Bangladesh", *CID Working Paper, Harvard* N° 28.

Anderson, K., and Pomfret, R., (2000). "Living Standard during Transition to a Market Economy: the Kyrgyz Republic in 1993 and 1996". *Journal of Comparative Economics*. N° 28, p. 502 – 523.

Ardeni, P. G., Andracchio, A., (2003). Consumption Poverty and Functioning Deprivation in Bosnia and Herzegovina". Paper presented at the Second International Conference of the Faculty of Economics *From Transition to Development*, Sarajevo October 9-11, 2003

Atkinson, A. B., *Poverty and Social Security*, Harvester Wheatsheaf, Hemel Hempstead, 1989.

Baulch, B. and McCulloch, N., (1998). "Being Poor and Becoming Poor: Poverty Status and Poverty Transitions in Rural Pakistan." *IDS Working Paper N*° 79. Institute for Development Studies: Sussex.

Bernabè, S. (2002). "Informal Employment in Countries in Transition: a Conceptual Framework". Centre for Analysis of Social Exclusion. London School of Economics.

BHAS, State Agency for Statistics et al., (2002). Welfare in Bosnia and Herzegovina, 2001: Measurement and Findings.

Bisogno M., and Chong, A., (2002), "Poverty and Inequality in Bosnia and Herzegovina after the Civil War", *World Development Report*, N° 30 (1), p. 61-75

Buchinsky, M. (1998), "Recent advances in quintile regression models: a practical guide for empirical research", *Journal of Human Resources*, 33(1): 88-126.

Chambers R., (1989). "Vulnerability, Coping and Policy"; Editorial Introduction to IDS Bulletin, Special Issue on "Vulnerability: How the Poor Cope", *IDS Bulletin* Vol.20, N°2; p. 1-7

Chaudhuri, S., Jalan, J., & Suryahadi, A. (2001). "Assessing Household Vulnerability to Poverty from Cross-sectional Data: A Methodology and Estimates for Indonesia". *Discussion Paper Series* no. 0102-52; Department of Economics, Columbia University.

Christiaensen, L., and H. Alderman. 2001. "Child Malnutrition in Ethiopia: Can Maternal Knowledge Augment the Role of Income." Africa Region Working Paper Series 22. World Bank, Washington, D.C.

Christiaensen, L., and Boisvert, R., (2000). "On Measuring Household Food Vulnerability: Case Evidence from Northern Mali". Working Paper 2000-05; Department of Applied Economics and Management; Cornell University.

Christiaensen, L., and Subbarao, K., (2001). "Towards an Understanding of Vulnerability in Rural Kenya". *mimeo*

Cuna and Alasia, (2003). "Territorial Variation in Poverty and Risk of Poverty in Bosnia and Herzegovina", paper presented at the International Conference "From Transition to Development", held in Sarajevo, October 2003.

Datt, Gaurav, et al. 1998. "Simulating poverty measures from regression models of household consumption." International Food Policy Research Institute, Washington D. C. Unpublished report.

Datt, G., Simler, K., Mukherjee, S., and Dava, G., (2000). "Determinants of Poverty in Mozambique: 1996-97". Research Report. International Food Policy Research Institute, Washington, D.C.

Datt, G., Joliffe, D., (1999). "Determinants of Poverty in Egypt: 1997". *Discussion Paper* N° 75 International Food Policy Research Institute, Washington, D.C.

Deaton, A., (1997). *The Analysis of Household Surveys: A Microeconometric Approach to Development Policy*. Baltimore, MD: Johns Hopkins University Press

Deaton, A. and J. Muellbauer, 'On Measuring Child Costs: With Applications to Poor Countries', *Journal of Political Economy*, Vol. 94, 1986, pp. 720–744.

Dercon, S. (2001). "Assessing Vulnerability to Poverty". Paper prepared for the Department for International Development (DfID), London.

Dercon, S., and Krishnan, P., (2000). "Vulnerability, seasonality, and poverty in Ethiopia". *Journal of Development Studies* Vol. 36 N°6: 25-53.

Djipa, D., Muzur, M., Lytle, P. F., (2002). "War-Torn Lives" in Voices of the Poor: From Many Land.

Efron, B., and R. Tibshirani (1993). *An Introduction to the Bootstrap*. New York: Chapman & Hall.

Foster J., Greer J., Thorbecke E., (1984), A class of decomposable poverty measures, Econometrica Vol. 52, N° 3.

Gardes, F., Langlois, S., Richaudeau, D., (1996). "Cross-section versus time series income elasticities of Canadian consumption". *Economic Letters*. N° 51, p. 169-175.

Glewwe, P. (1991). "Investigating the determinants of household welfare in the Côte d'Ivoire." *Journal of Development Economics* 35: 307-338.

Glewwe P., Hall G., (1998) "Are Some Groups More Vulnerable to Macroeconomic Shocks than Others? Hypothesis Test based on Panel Data from Peru", *Journal of Development Economics*, Vol. 56.

Grootaert, C., and Braithwaite, J.,(1998). Poverty correlates and indicator based targeting in Eastern Europe and the Former Soviet Union. *Policy Research Working Paper* N° 1942, The World Bank, Washington DC.

Haughton, J., Haughton, D., Loan, L. T., Phong, N., (2001). "Shooting Stars and Sinking Stones", *mimeo*.

Hoddinott, J., and Quisumbing, A., (2003). "Methods for Microeconometric Risk and Vulnerability Assessments". Social Protection Discussion Paper Series, N° 0324. The World Bank

Holzman, R., and S. Jorgensen. 2000. "Social risk management: A new conceptual framework for social protection and beyond". *Social Protection Discussion Paper* N° 0006. World Bank. Washington, D.C.

Houthakker, H., (1965). "New Evidence on demand elasticities". *Econometrica*, N° 33, p. 277-288.

Judge, G., Griffiths, W., Hill, R., Lutkepohl, H., Lee, T., (1985). *The Theory and Practice of Econometrics*. 2nd Edition. Wiley and Sons. New York.

Koenker, R., and Basset, G., (1978). "Regression Quantile". Econometrica, N° 46, p.33-50.

Kuh, E., (1959). "The validity of cross section estimated behaviour equations in time series applications". *Econometrica*. N° 27, p. 197 – 214.

Lanjouw, P., Lanjouw, J., and Elbers, C. (2002) "Micro-Level Estimation of Welfare," Policy Research Working Paper Series 2911, The World Bank.

LSMS, (2001). "Basic Information Document - Bosnia and Herzegovina".

Midhet, F., Becker, S., and Berendes, H., (1998). "Contextual determinants of maternal mortality in rural Pakistan." *Social Science and Medicine* Vol. 46, N° 12. p.1587–98.

Moser, C., (1996) 'Confronting Crisis: A Comparative Study of Household Responses to Poverty and Vulnerability in Four Poor Urban Communities''. *Environmental Sustainable Development Studies Series* N° 8, The World Bank. Washington, D.C.

Moser, C., (1998). "The Asset Vulnerability Framework: Reassessing Urban Poverty Reduction Strategies." *World Development*. Vol. 26, N° 1. p. 1-19.

Mukherjee, S., and Benson, T., (2003). "The Determinants of Poverty in Malawi, 1998", *World Development*, Vol. 31 N° 2, p. 339 – 358.

Morduch, J, and Gamanou, G. (2002). "Measuring Vulnerability to Poverty". Discussion Paper No. 2002/58, World Institute for Development Economics Research (WIDER); United Nations University.

PRSP, (2002). Poverty Profile in Bosnia and Herzegovina. PRPS Team

Prichett, L., Suryahadi, A., and S. Sumarto. 2000. "Quantifying Vulnerability to Poverty: A Proposed Measure, Applied to Indonesia." *Policy Research Working Paper* N° 2437. The World Bank: Washington, D.C.

Pullan, S., (1999), "New Approaches to Poverty and New Forms of Institutional Charity in Late Medieval and Renaissance Italy" in V. Zamagni (1999), *Povertà e Innovazioni Istituzionali in Italia*. Il Mulino

Ravallion, M., and Wodon, Q., (1997). "Poor Areas, or Only Poor People?" *Journal of Regional Science*, 39(4): 689–711.

Ricci, G., (1999). "Nel paese di Anomalia (vergognosi/declassati)", in V. Zamagni (1999), *Povertà e Innovazioni Istituzionali in Italia*. Il Mulino, Bologna.

Rukanesen, K., (2003) "Female Headed Households in Bosnia and Herzegovina", United Nations in Bosnia and Herzegovina. UNDP

Sen, A., (1993). *Capabilities and Wellbeing*. in Nussbaum, M.C., and Sen, A., *The Quality of Life*, Clarendon Press, Oxford

Shaffer, P., (2001). "New Thinking on Poverty: Implications for Poverty Reduction Strategies" CIS – University of Toronto, UNDESA.

Sinha, S., and Lipton, M., (1999). "Damaging Fluctuations, Risk and Poverty: A Review", Poverty Research Unit at Sussex (PRUS), University of Sussex, Brighton

Skoufias, E., Davis, B., and Behrman, J., (1999). An evaluation of the selection of beneficiary households in the Education, Health and Nutrition Programme (PROGRESA) of Mexico. International Food Policy Research Institute, Washington DC.

UNDP, (2002). *Human Development Report Bosnia and Herzegovina 2002*. United Nations Development Programme Bosnia and Herzegovina.

UNDP (2003). Achieving Millennium Development Goals in Bosnia and Herzegovina-Human Development Report 2003. United Nations Development Programme.

Vodopivec, M., (2002). Labour Market in the Post-War Bosnia and Herzegovina: How to Encourage Business to Create Jobs and Increase Worker Mobility. Human Development Unit – South East Europe Country Unit Europe and Central Asia Region. The World Bank.

Wodon, Q., (2003). "SimSIP: Simulations for Social Indicators and Poverty." World Bank Technical Paper. World Bank, Washington, D.C.

World Bank (2002). Bosnia and Herzegovina: from Aid Dependency to Fiscal Self Reliance. A Public Expenditure and Institutional Review. Poverty Reduction and Economic Management Unit. Europe and Central Asia Region.

World Bank, (2003). Bosnia and Herzegovina Poverty Assessment. Poverty Reduction and Economic Management Unit. Europe and Central Asia Region. The World Bank.

Wu, C.F.J., (1986). Jackknife, bootstrap and other resampling methods in regression analysis. *Annals of Statistics* Vol. 14, pp. 1261-1350.